Author’s response to reviews

Title: Enhancement of hepatitis virus immunoassay outcome predictions in imbalanced routine pathology data by data balancing and feature selection before the application of support vector machines

Authors:
Alice Richardson (alice.richardson@anu.edu.au)
Brett Lidbury (brett.lidbury@anu.edu.au)

Version: 2 Date: 19 Mar 2017

Author’s response to reviews:

Thank you for the opportunity to tidy up this submission before it is sent to peer review. The Editor's comments, and our responses, are given below.

Editor: The authors should check the manuscript before being sent to reviewers. In particular: On page 5, the sentence ending with reference [14] is not complete, and the reference is missing.

Response: The reference has been added, and the sentence on page 5 completed.

Editor: Add more information on the dataset in the abstract.

Response: This has been done.

Editor: Please carefully check English and grammar throughout the paper, being careful about the use of the word "very" (also in the abstract).

Response: This has been done. The single use of the word “very”, in the abstract, has been removed.